Author’s response to reviews

Title: Effect of a Postpartum Family Planning Intervention on Postpartum Intrauterine Device Counseling and Choice: Evidence from a Cluster-Randomized Trial in Tanzania

Authors:

Erin Pearson (pearsone@ipas.org)
Leigh Senderowicz (senderowicz@wisc.edu)
Elina Pradhan (epradhan@worldbank.org)
Joel Francis (joel.francis@wits.ac.za)
Projestine Muganyizi (promuga@yahoo.com)
Iqbal Shah (ishah@hsph.harvard.edu)
David Canning (dcanning@hsph.harvard.edu)
Mahesh Karra (mvkarra@bu.edu)
Nzovu Ulenga (nulenga@mdh-tz.org)
Till Bärnighausen (till.baernighausen@uni-heidelberg.de)

Version: 2 Date: 13 Apr 2020

Author’s response to reviews:

Author Responses to Editor Comments: BMWH-D-19-00770

Authors: We thank the reviewers for their final review and the Assistant Editor for their comments and have provided a point-by-point response below.

Assistant Editor

1. Title Page
--On your title page, please list the full institutional email addresses for all authors. The corresponding author should be indicated.
• Authors: We have added the institutional email addresses for each author. The corresponding author (Erin Pearson) is indicated using an asterisk.

2. Ethics Approval and Consent to Participate
--In your ‘Ethics Approval and Consent to Participate’ section, please confirm whether informed consent obtained was written or verbal. If verbal, please state the reason and whether the ethics committee approved this procedure.
• Authors: We have specified in this section that informed consent was written.
3. Authors' Contributions
--We have noted that authors [Projestine Muganyizi, PM], [Nzovu Ulenga, NU] and [Till Bärnighausen, TB] are missing in the listed authors' contributions. The individual contributions of ALL authors to the manuscript should be specified in the Authors’ Contributions section. Guidance and criteria for authorship can be found here:
http://www.biomedcentral.com/submissions/editorial-policies#authorship
• Authors: These three authors were included under the sentences specifying contributions of all authors. We have made this clearer by listing all authors in each of these sentences rather than simply stating “all authors contributed”.

4. Clean Manuscript
--At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files. Please ensure that all figures, tables and additional/supplementary files are cited within the text.
• Authors: We have saved and uploaded a clean version of all files.